Author's response to reviews

Title: Effectiveness of a Combined Strategy to Improve Therapeutic Compliance and Degree of Control Among Patients with Hypercholesterolaemia: a Randomised Clinical Trial

Authors:

Ignacio Párraga-Martínez (iparraga@sescam.jccm.es)
Joseba Rabanales-Sotos (jrabanales@icmm.es)
Fernando Lago-Deibe (Fernadolsidro.Lago.Deibe@sergas.es)
Juan M Téllez-Lapeira (juanmtellez@gmail.com)
Francisco Escobar-Rabadán (fjescobarr@sescam.jccm.es)
Alejandro Villena-Ferrer (alejandrov@sescam.jccm.es)
Mariano Blasco-Valle (mblascov@salud.aragon.es)
José M Ferreras-Amez (chemaferreiras@hotmail.com)
Susana Morena-Rayo (smrsusana@yahoo.es)
José M del Campo-del Campo (jmariadc@sescam.jccm.es)
Maria C Ayuso-Raya (candeayuso@hotmail.com)
José J Pérez-Pascual (josejorge24@yahoo.es)

Version: 2 Date: 14 December 2014

Author's response to reviews: see over
The Editor,
BMC PUBLIC HEALTH

Dear sir/madam,

I enclose the manuscript entitled, "Effectiveness of a combined strategy to improve therapeutic compliance and degree of control among patients with hypercholesterolaemia: a randomised clinical trial." for publication in the study protocol section of BMC PUBLIC HEALTH.

This is an original study that has not been previously published and is not under consideration for publication elsewhere. All authors have read and approved the manuscript, contributed intellectually to the study and have no conflicts of interest.

Informed consent was obtained from all the participants.

This study was funded by the Carlos III Institute of Health (Instituto de Salud Carlos III/ISCiii), a public research body which comes under the aegis of the Ministry of Economic Affairs & Competitiveness but is functionally part of the Ministry of Health, Social Services & Equality. Co-funding was obtained from the EU European Regional Development Fund (ERDF). Funding was obtained through a Health Research Project Subprogramme call for funding requests (dossier no. PI12/01955): resolution dated 20 December 2012.

I should like to thank you in advance for your time and attention in this matter, and look forward to hearing from you.

Yours faithfully,

-------------------------------------------------

Ignacio Párraga Martínez
Specialist in Family and Community Medicine, Castile-La Mancha Health Service (Servicio de Salud de Castilla-La Mancha/SESCAM).